Author's response to reviews

Title: A Supratentorial Primitive Neuroectodermal Tumor presenting with Intracranial Hemorrhage in a 42-year-old male: Case Presentation and Literature Review

Authors:

Evangelos K Papadopoulos MD (neuroevan@yahoo.gr)
Kostas N Fountas MD, PhD (fountas@med.uth.gr)
Alexandros G Brotis MD, PhD (alexgbrodis@yahoo.com)
Konstantinos N Paterakis MD, PhD (kpaterakis@yahoo.com)

Version: 3 Date: 4 December 2012

Author's response to reviews: see over
Dear Editor,

Please consider for publication at your journal our revised original manuscript entitled: “A Supratentorial Primitive Neuroectodermal Tumor presenting with Intracranial Hemorrhage in a 42-year-old male: Case Presentation and Literature Review”.

The requested changes on the manuscript format have been performed.

More specifically:

1. The word “Report” has been substituted for “Presentation” in the title.
2. The acronym “PNET” has been defined at the abstract section.
3. Unjustified conclusions regarding the Ki-67 index have been removed from the abstract section.
4. The keywords have been transferred after the abstract section.
5. The acronym ICP has been added on page 3.
6. The word “Report” has been substituted for “Presentation” on page 3.
7. The word “report has been substituted for “communication” on page 3.
8. The acronym MRI has been added on page 6.
9. Unjustified conclusions regarding the Ki-67 index have been removed from the conclusions section.
10. Correction on the source of the informed consent source (relatives instead of the patient).
11. Language improvement has taken place.
12. All changes have been marked using the Word’s “Track changes” application and are highlighted and underlined.
13. Major changes are described in the present submission cover letter as requested.

Thanks in advance for your prompt attention and looking forward for your response.

Sincerely

Kostas N. Paterakis
Assistant Professor of Neurosurgery